

ORCA - Online Research @ Cardiff

This is an Open Access document downloaded from ORCA, Cardiff University's institutional repository:https://orca.cardiff.ac.uk/id/eprint/132543/

This is the author's version of a work that was submitted to / accepted for publication.

Citation for final published version:

Shepherd, Victoria 2020. Advances and challenges in conducting ethical trials involving populations lacking capacity to consent: a decade in review. Contemporary Clinical Trials 95, 106054. 10.1016/j.cct.2020.106054

Publishers page: http://dx.doi.org/10.1016/j.cct.2020.106054

Please note:

Changes made as a result of publishing processes such as copy-editing, formatting and page numbers may not be reflected in this version. For the definitive version of this publication, please refer to the published source. You are advised to consult the publisher's version if you wish to cite this paper.

This version is being made available in accordance with publisher policies. See http://orca.cf.ac.uk/policies.html for usage policies. Copyright and moral rights for publications made available in ORCA are retained by the copyright holders.



Advances and challenges in conducting ethical trials involving populations lacking capacity to consent: a decade in review

Dr Victoria Shepherd

Centre for Trials Research, Cardiff University, UK

Corresponding author:

Dr Victoria Shepherd

Centre for Trials Research, 4th floor Neuadd Meirionnydd, Heath Park,

Cardiff CF14 4YS

ShepherdVL1@cardiff.ac.uk

Abstract

Informed consent is an essential requirement prior to clinical trial participation, however some 'vulnerable' groups, such as people with cognitive impairments and those in medical emergency situations, may lack decisional capacity to consent. This raises ethical and practical challenges when designing and conducting clinical trials involving these populations, who are frequently excluded as a result. Despite recent advances in improving informed consent processes, there has been far less attention paid to the enrolment of adults lacking capacity.

Exclusion criteria are an important determinant of the external validity of clinical trial results. The exclusion of these populations, and consent-based recruitment biases which arise from the challenges of identifying and involving surrogate decision-makers, leads to trials which are not representative of the clinical population.

This article discusses the involvement of adults who lack decisional capacity to consent in clinical trials and presents the advances over the previous decade and the remaining ethical challenges for the inclusion of this under-represented population in research.

Keywords

Informed consent, clinical trials, ethics, surrogate decision-making

Introduction

Despite the growing focus on improving informed consent processes for clinical trials over the previous decade [1], enrolment of participants who lack decisional capacity to provide informed consent has been the subject of much less attention. This is due, in part, to the challenging ethical issues that surround such trials, and uncertainty about how they should be addressed. However, this issue will become even more pressing in the coming decades with an increase in the prevalence of neurodegenerative and other conditions that can affect the ability to make informed decisions about participation in clinical trials [2]. In addition, the current coronavirus pandemic has resulted in a surge of critically ill patients and a corresponding explosion of clinical trials evaluating therapeutic agents for COVID-19 [3]. Patients admitted to critical care units are frequently unable to provide informed consent [4], and COVID-19 itself is associated with a number of psychiatric and neuropsychiatric effects, including delirium which is common in the acute stage [5]. Thus the need to conduct clinical trials to inform and transform evidence-based care for these growing populations is intensifying.

Over the previous decade there has also been a growing awareness of the issue of under-represented and underserved populations in research. This recognises that there is a considerable difference between trial and clinical populations, where participants in trials do not represent the characteristics of patients who could benefit from the drugs in 'real world' clinical practice settings. This includes that trial populations are younger [6] less representative of women and minority groups [7] and have fewer health conditions [8] than those in the affected population. Approximately 58% of the total US population are typically excluded from research studies [9], and a third of randomised controlled trials are at high risk of bias, most commonly because the clinical population used is not appropriate for the trial [8]. Exclusions of people with cognitive impairment are seen in areas such as geriatrics research [10], rehabilitation interventions after hip fracture [11], learning disabilities [12], peri-operative medicine [13], trauma [14] and neurological research [15]. Clinical trials in other areas such as emergency research also encounter significant ethical challenges around informed consent which impacts on the populations included in trials [16]. Even in trials that are designed to include adults who lack capacity to consent, the proportion and numbers of participants lacking capacity actually recruited are worryingly small [17]. This recruitment bias leads to underpowered or abandoned studies [15] and trials which are not representative of the clinical population [14, 16, 18], resulting in evidence 'biased' medicine for these populations [19].

The prevalence of incapacity in different settings and populations can be hard to determine, in part due to the decision-specific nature of capacity, however a systematic review reported that the proportion of patients with incapacity in medical settings was 34% (95% CI 25–44%) and 45% (95% confidence interval (CI) 39–51%) in psychiatric settings [20]. Research conducted in UK care homes (long term care facilities) with older people found 71% of participants lacked capacity to consent [21], this proportion rises to over 90% in critical care settings [4]. Decisional capacity to consent to research is generally accepted to be the ability of a potential research subject to understand and process the information that is necessary to make an informed decision regarding study participation [22] and to communicate that decision [23]. Decisional capacity, however, is task-specific and not a static concept. People assessed as lacking capacity may have capacity to make some decisions but not others, may regain and lose capacity over time, or experience fluctuating capacity. Trials involving adults with cognitive impairment raises ethical issues, primarily because these groups are unable to provide adequate informed consent or protect their own interests [22]. National research regulations and ethical guidance require that investigators provide additional safeguards for vulnerable subjects who participate in research [24]. These regulations were originally designed to protect 'vulnerable'

research participants. However this exclusionary ethics puts vulnerable populations at risk, while trying to protect them from exploitation [14]. More recent changes in the attitudes of research communities' are starting to recognise that these groups should be protected *through* research, rather than *from* research [9]. This is echoed in international ethical guidance which states that adults who are not capable of giving informed consent must be included in health-related research unless a good scientific reason justifies their exclusion [25]. Incoming European regulations require that clinical trials participants should represent the population groups, for example gender and age groups, that are likely to use the medicinal product investigated in the clinical trial [26], although only if the trial cannot be conducted equally well with participants who are able to provide consent, and the research addresses health needs relevant to people with those particular conditions [25].

Eligibility criteria are intended to be used to recruit participants that are representative of the patient populations who will ultimately receive the medication or intervention in clinical practice [27]. Exclusion criteria are commonly used in trials to ensure the safety and protection of participants; however, some use a very restricted profile for patient eligibility [27]. Eligibility criteria which exclude groups without valid reasons can result in research samples that do not represent the diversity, symptom complexity, or daily challenges of the clinical population [15] which negatively affects the external validity of the trial [27]. Excluding patients with cognitive impairments is common but a robust rationale is rarely provided or discussed as a potential limitation [10, 15]. Arbitrary use of this criterion excludes large numbers of the clinical population and so limits the pool of potential participants, and, as these excluded individuals are disproportionately from vulnerable populations, this exclusion also raises ethical concerns [15]. Current exclusion criteria, ethical barriers, and enrolment practices all impact on trial populations and so may reduce the clinical utility of research findings, leading to inequity in evidence-informed interventions for these populations [4, 10, 11, 28]. While there have been calls for more active involvement of research funding organisations in the scrutiny of the justifications behind this exclusion, such as reviewing eligibility criteria [28], the clinical trials community can play a large role in designing more inclusive research, and identifying and addressing the ethical challenges that impact on the conduct of ethical trials with populations lacking capacity to consent. The start of a new decade presents an opportunity to pause and review the current issues encountered when conducting trials involving adults who lack decisional capacity to consent. This article discusses a number of the barriers and presents recent advances and remaining challenges for the inclusion of this under-represented population. Recommendations are made that seek to address these challenges for future clinical trials.

Challenges of surrogate consent

For adults who lack capacity to consent to participate in a trial, the consent or agreement of a family member or other person with a close personal relationship with the person concerned must be sought prior to their enrolment [25]. Surrogate or proxy decision-makers must evaluate to what extent trial participation is consistent with the person's wishes and values, and any previously stated preferences regarding the person's willingness to enrol in research should be respected [25]. In situations where a family member or friend is not available, in some jurisdictions researchers may obtain the permission of an alternative representative [25] who may be an independent member of the healthcare team or an individual appointed by the state. However, a number of challenges may be encountered, including problems with gaining ethical approvals, identifying surrogates, and the problems faced by surrogates making a decision about participation on behalf of another including determining the preferences and values of the person they represent.

Complexities of legislative frameworks and ethical review processes

The legal provisions for the involvement of adults who lack capacity to consent, and alternative arrangements for their inclusion in clinical trials, varies between different legal jurisdictions. One study which examined the ethical approval process for an intensive care observational study found substantial and persisting variations between EU member states in the organisation, structures, processes, efficiency, and decision-making of Research Ethics Committees (RECs) [29]. Within the US, there is considerable variability between Institutional Review Board (IRB) practices on surrogate consent [30]. However, most US states do not have laws regarding surrogate consent for research [31], and reliance on healthcare proxies is problematic because many people have no legally authorized representative for medical decision-making [32]. The authors of a recent study examining these practices argue that this variability may have adverse consequences for the much needed research involving adults who lack capacity, and call for guidance to help clarify current regulations [30]. A similar 'patchwork of laws' exist in many other countries, such as Australia [33]. These legislative complexities and accompanying heterogeneity in ethical approvals processes are challenging when conducting international clinical trials involving adults who lack capacity. The importance of international co-ordination of COVID-19 trials of candidate drugs and vaccines has been highlighted, including the need to ensure the representativeness of the trial population through the inclusion of vulnerable groups [34].

Research has identified the impact of the complex legislation governing research involving adults lacking capacity to consent in the UK [19]. This complexity leads to misunderstanding and misinterpretation of the legislation by researchers who design trials involving adults who lack capacity [35], RECs who review these trials [36] and healthcare professionals caring for these populations [37]. Current ethical guidelines fail to support the inclusion of participants unable to consent through the lack of specific and appropriate advice, which is in itself ethically problematic [14]. Further guidance is urgently needed for RECs/IRBs, the trials community, and healthcare professionals in order to improve confidence and reduce inconsistencies that act as barriers to conducting trials involving adults lacking capacity to consent.

Concerns about surrogate decision-making

Another issue that has been the focus of attention over the previous decades is surrogate decisionmaking. This includes aspects of the patient-surrogate relationship, concerns about 'accuracy' of the surrogate's decision, and how the risks and benefits of participation are balanced alongside respecting the preferences and values of the person who lacks capacity. Numerous studies have explored the accuracy of a surrogate's substituted judgement, often presenting a hypothetical study to a patient and surrogate dyad and asking the surrogate to predict what the patient would decide about participating [38-43]. Across these studies, accuracy varied with a patient-surrogate discrepancy reported as between 32% and 42% depending on the hypothetical scenario in one study [40], and an accuracy rate of 76% reported in another study [39]. Newman and colleagues found that the overall percentage of discrepancy increased as the perceived risk associated with the hypothetical study rose [42]. However, it is important to note that such studies have important limitations and methodological flaws [44]. This includes that they are based on the assumption that decision-making in hypothetical and actual situations are the same [45], and that the patient's own 'decision' should be seen as the gold standard, although they too are expressing a prediction which may not be accurate [46]. In a number of studies, potential participants asked about future research participation in the event of incapacity reported that the choice of who acted as their proxy was important [47-49] which wasn't necessarily the family member accompanying them to clinic (as used in these dyad accuracy studies).

Patients are willing to grant leeway to their surrogates when making decisions about research on their behalf on the basis that these are contextual decisions and many relevant factors won't be known in advance [50].

The discordance between patient and surrogate decisions may also be due to a difference in motivation for participation. Critical care patients' motives are primarily altruistic and centre on a desire to help others and advance science [4], and whilst this is also a consideration for their surrogates' [51], surrogates' motives focus on a belief that their loved one could benefit from participation [4]. Surrogates themselves describe how they balance a number of different factors during the decision-making process, including attempting to honour the wishes of the person they represent whilst assessing the risks and benefits of participation [52]. Despite concerns from some quarters about this lack of 'accuracy', many studies report that patients and the public want their surrogate to be involved in decisions about research participation [53–55] and that surrogates want to be involved [46, 56].

Addressing the challenges of surrogate decision-making

One of the challenges in surrogate consent is identifying a surrogate who is willing and able to make a decision about trial participation. Previous studies have found that 40% - 80% of eligible critical care patients who lacked decision-making capacity were not enrolled in a study because no family member was available to provide surrogate consent [57, 58], with 15% having no visitors at any time during their stay [59]. Similar issues identifying, contacting, and receiving a reply from surrogates are encountered in trials in other setting such as care homes [60]. Additional challenges may be encountered during public health emergencies, such as COVID-19, where family members may not be permitted to visit patients or may themselves be unwell. Alternative arrangements are in place, such as conducting discussions with surrogates by phone or video conference and obtaining consent via electronic or other means [61], and may be useful to continue beyond the current pandemic.

One approach that may address this challenge is to encourage patients to prospectively appointment or nominate a surrogate for research, although it may not be helpful for those considered 'unbefriended' patients (those without any family member or friend who can act as a surrogate decisions-maker). When surveyed about who should be involved in decisions about their participation in research during periods of incapacity, most older people (88%) preferred the person they appointed as their surrogate decision-maker for healthcare matters [62], and 96% were willing to appoint a surrogate for research [63]. Generally in the US, a surrogate may be able to provide consent through their appointment as legally authorized representative, although the situation regarding treatment and research decisions across the different states is very complex [31]. In England and Wales, despite the growing focus on advance planning in the event of incapacity, there is no legal mechanism to appoint a surrogate for research [64], although someone can be appointed to make decisions about health and welfare using Power of Attorney [23]. Further research is needed to explore whether expanding and harmonising legal provisions for prospectively appointing a surrogate for research does address the challenge of identifying a surrogate.

Even once identified, surrogates can find being involved in making decisions about research participation overwhelming [65], with nearly all proxies experiencing some degree of burden [66]. In one study of family members of critically ill patients, being asked to provide consent for research was associated with post-traumatic stress symptoms in 35% of family members interviewed, compared with less than 10% in those involved in 'everyday' decisions about clinical care [67]. However, despite the number of interventions developed to improve informed consent, very few interventions have been developed to improve surrogate decision-making for participating in trials. The interventions

that have been developed either focus on elements of informedness, such as a computer-based education module to improve surrogates' understanding of the process of informed consent [68] or on enhancing surrogate decisions such as a decision aid to support surrogate decision-making [69]. Further research to explore patients and surrogates' views and experiences of surrogate decision-making about clinical trial participation is needed, alongside interventions to address the challenges and burdens experienced by surrogates.

Another approach to reducing the uncertainty for surrogates and protecting the autonomy of patients through ensuring surrogate decisions are more representative of their preferences and values, is by the use of advance research directives (ARD). An ARD can be used to document a person's wishes concerning research participation in the event of a loss of capacity, and also specify the general risk level that they are prepared to accept [70]. ARDs could be particularly useful for people who are living with a condition that may cause cognitive abilities to decline in the future, such as Parkinson's disease or dementia [70]. Although not necessarily legally binding, if participants have made advance directives for participation in research while fully capable of giving informed consent, the directives should be respected [25]. While many countries have recognized the importance and need for ARDs as a tool for research, particularly in dementia, clear legal regulations are still lacking in most countries [71]. Enabling the creation of ARDs may encourage discussion about future wishes between the patient and their surrogate and so enable surrogate decisions about research to be made that are in accordance with the person's preferences and wishes.

Informed consent in emergency research

It is widely accepted that it is impossible to obtain sufficiently informed consent in emergency and prehospital research, due in part to the physical and psychological effects of the medical emergency, but also the limited time frames for the initiation of the treatment or intervention being tested [72, 73]. However, the requirement in some jurisdictions to obtain consent from either the patient or (if legislation permits) their surrogate prior to enrolment in a trial is a substantial barrier to emergency research [74]. Acute and critical care clinical trials have a four-fold higher risk of discontinuation due to recruitment issues [75].

Impact of requiring patient or surrogate consent

The consent mechanism used for an emergency trial has a profound effect on the ability to conduct the trial and the population enrolled [16]. It has been estimated that a trauma study requiring recruitment within a few hours of admission will see a 80% to 90% reduction in the number of potentially eligible patients if consent from either the patient or their legally authorised representative (LAR) is required [16]. In an observational study of 1,734 trauma patients in the US, the requirement for consent from either the patient or their LAR also led to recruitment bias as patients who were female, had blunt trauma, or were transported a shorter distance were significantly more likely to have LARs [74]. This consent-based recruitment bias was also seen in the CONTROL trial of factor VIIa in acute haemorrhagic shock, where the requirement for either individual or LAR consent meant that US participants bled more slowly, were randomized later, and represented a less injured population than those eligible but not enrolled [16]. The consent requirements also impacted on the numbers recruited as only 17% of all US eligible patients were enrolled, largely because of the delayed arrival of LARs with 63% arriving at the hospital >12 h after the patient and so missing the inclusion criteria [16]. The requirement for these 'consent rituals' may also lead to harm through delaying potentially

life-saving treatment when such delays would not be encountered in clinical practice, and potentially reduce or obscure any treatment effect [76].

Alternative models of consent in emergency research

In emergency situations, when prior consent of the patient is not possible, and the subject's legally acceptable representative is not available, enrolment of the patient should follow alternative procedures described in the protocol that has been reviewed and approved by the IRB/REC. In many jurisdictions, such as the US, EU, Canada, and Australia regulations allow research in emergency situations to be conducted without prospective informed consent and to use alternative models of consent provided certain conditions are met [77]. The subject or the patient's legally acceptable representative should be informed about the trial as soon as possible and consent to continue and other consent as appropriate should be requested [78]. In the US, an a Exception For Informed Consent (EFIC) may be allowed, provided certain conditions are met such as community consultation with members of the community in which the research will be conducted [79]. A waiver of informed consent (WIC) regulations apply to other types of research [80].

In the EU, the Clinical Trial Directive introduced in 2001 [81] appeared to overlook critical care and emergency research [82], which was subsequently covered by the introduction of national legislation in many EU member states including the UK [83]. Although a lack of harmonisation means that deferred consent (consent sought from either the patient and/or surrogate at a later timepoint) is accepted in acute emergency research in only half of the member states [84]. Amongst other reasons, this complexity has led to a reduction in clinical trial activity in Europe, particularly in academic trials [82]. Although it is hoped that the new incoming Clinical Trials Regulation [26] will facilitate and harmonise emergency research across the EU [85], the impact has yet to be seen, and the position in the UK once it leaves the EU remains unclear.

Addressing the challenges of deferred or waiver of consent

Gaining a favourable ethical opinion and regulatory approvals for conducting a trial using deferred consent or EFIC can be very challenging, particularly those in the 'middle ground' where a patient may be awake or coherent, yet extreme pain or a lack of time allows no possibility of prospective informed consent [86]. A review of US studies using EFIC/WIC over the past 20 years identified only 28 studies, with the authors suggesting that more studies using EFIC/WIC are needed and a greater description should be published of both the justification for the use of EFIC/WIC in these trials and the process followed to ensure transparency [80]. A useful approach has been suggested for those who design and review studies using deferred consent which proposes a practical framework of 'questions and considerations' that a researcher should be able to answer if planning to apply to a REC for waiver of consent in the context of emergency care research [87]. Further practical and evidence-based approaches are needed in order to improve the design, review and regulation processes of trials in emergency settings, and to prevent and resolve slow patient recruitment in randomized clinical trials conducted in the critical and emergency care setting [75].

Conclusion

Trials involving adults who lack capacity to consent are methodologically and ethically challenging. These challenges impact on the ability of trials to recruit sufficient number of participants, and consent-based recruitment biases raise questions about the external validity of the results. With

changing demographics which are predicted to increase the numbers of adults lacking decisional capacity, and the current global COVID-19 pandemic, further research and resources are needed to help address these challenges and to support the inclusion of people with impaired decision-making capacity.

Guidance is needed for those designing clinical trials which provides simple and accessible information and resources to help them understand and navigate the complex legal and ethical requirements of surrogate consent and deferred consent. Further research is needed to explore clarifying and extending frameworks for advance planning for research participation, such as prospectively appointing a surrogate for research decisions and creating Advance Research Directives to document preferences about future research participation.

These measures should be complemented by the development of evidence-based strategies and interventions to inform and support substitute decision-makers across different research settings. Greater transparency and reporting of the use of alternative models of consent such as deferred consent and EFIC/WIC should be encouraged, which in turn will support the development of practical and evidence-based approaches to improve the design, review, and regulation of trials in emergency settings. A wider review of the ethical barriers and a more active involvement of research funding organisations and RECs/IRBs to scrutinise the justifications behind the exclusion of adults lacking capacity might be also be warranted.

Acknowledgements

This paper was supported by a Wellcome Trust ISSF3 Consolidator Grant from Cardiff University. The research was funded by a National Institute of Health Research Doctoral Research Fellowship (NIHR-FS-2016), funded by the Welsh Government through Health and Care Research Wales.

References

- 1. Lentz J, Kennett M, Perlmutter J, Forrest A. Paving the way to a more effective informed consent process: Recommendations from the Clinical Trials Transformation Initiative. Contemporary Clinical Trials. 2016;49:65–9.
- 2. Dementia. World Health Organization. https://www.who.int/news-room/fact-sheets/detail/dementia. Accessed 24 Feb 2020.
- 3. Phua J, Weng L, Ling L, Egi M, Lim C-M, Divatia JV. Intensive care management of coronavirus disease 2019 (COVID-19): challenges and recommendations. The Lancet Respiratory Medicine. 2020;8:506–17.
- 4. Burns KEA, Zubrinich C, Tan W, Raptis S, Xiong W, Smith O, et al. Research recruitment practices and critically ill patients. A multicenter, cross-sectional study (the Consent Study). Am J Respir Crit Care Med. 2013;187:1212–8.
- 5. Rogers JP, Chesney E, Oliver D, Pollak TA, McGuire P, Fusar-Poli P, et al. Psychiatric and neuropsychiatric presentations associated with severe coronavirus infections: a systematic review and meta-analysis with comparison to the COVID-19 pandemic. The Lancet Psychiatry. 2020;:S2215036620302030.
- 6. Banzi R, Camaioni P, Tettamanti M, Bertele' V, Lucca U. Older patients are still under-represented in clinical trials of Alzheimer's disease. Alzheimers Res Ther. 2016;8. doi:10.1186/s13195-016-0201-2.
- 7. Heller C, Balls-Berry JE, Nery JD, Erwin PJ, Littleton D, Kim M, et al. Strategies Addressing Barriers to Clinical Trial Enrollment of Underrepresented Populations: a Systematic Review. Contemp Clin Trials. 2014;39:169–82.
- 8. He J, Morales DR, Guthrie B. Exclusion rates in randomized controlled trials of treatments for physical conditions: a systematic review. Trials. 2020;21. doi:10.1186/s13063-020-4139-0.
- 9. Spong CY, Bianchi DW. Improving Public Health Requires Inclusion of Underrepresented Populations in Research. JAMA. 2018;319:337.
- 10. Taylor JS, DeMers SM, Vig EK, Borson S. The Disappearing Subject: Exclusion of People with Cognitive Impairment and Dementia from Geriatrics Research. Journal of the American Geriatrics Society. 2012;60:413–9.
- 11. Sheehan KJ, Fitzgerald L, Hatherley S, Potter C, Ayis S, Martin FC, et al. Inequity in rehabilitation interventions after hip fracture: a systematic review. Age Ageing. 2019;48:489–97.
- 12. Feldman MA, Bosett J, Collet C, Burnham-Riosa P. Where are persons with intellectual disabilities in medical research? A survey of published clinical trials. Journal of Intellectual Disability Research. 2014;58:800–9.
- 13. Lindsay WA, Murphy MM, Almghairbi DS, Moppett IK. Age, sex, race and ethnicity representativeness of randomised controlled trials in peri-operative medicine. Anaesthesia. n/a n/a. doi:10.1111/anae.14967.

- 14. Jensen JS, Reiter-Theil S, Celio DA, Jakob M, Vach W, Saxer FJ. Handling of informed consent and patient inclusion in research with geriatric trauma patients a matter of protection or disrespect? Clin Interv Aging. 2019;14:321–34.
- 15. Trivedi RB, Humphreys K. Participant exclusion criteria in treatment research on neurological disorders: Are unrepresentative study samples problematic? Contemporary Clinical Trials. 2015;44:20–5.
- 16. Holcomb JB, Weiskopf R, Champion H, Gould SA, Sauer RM, Brasel K, et al. Challenges to Effective Research in Acute Trauma Resuscitation: Consent and Endpoints. Shock. 2011;35:107–113.
- 17. Shepherd V, Wood F, Griffith R, Sheehan M, Hood K. Protection by Exclusion? The (lack of) inclusion of adults who lack capacity to consent to research in clinical trials in the UK. Trials. 2019.
- 18. Zimmermann JB, Horscht JJ, Weigand MA, Bruckner T, Martin EO, Hoppe-Tichy T, et al. Patients enrolled in randomised clinical trials are not representative of critically ill patients in clinical practice: Observational study focus on tigecycline. International Journal of Antimicrobial Agents. 2013;42:436–42.
- 19. Shepherd V. Research involving adults lacking capacity to consent: the impact of research regulation on "evidence biased" medicine. BMC Medical Ethics. 2016;17:8.
- 20. Lepping P, Stanly T, Turner J. Systematic review on the prevalence of lack of capacity in medical and psychiatric settings. Clinical Medicine. 2015;15:337–43.
- 21. Hood K, Nuttall J, Gillespie D, Shepherd V, Wood F, Duncan D, et al. Probiotics for Antibiotic-Associated Diarrhoea (PAAD): a prospective observational study of antibiotic-associated diarrhoea (including Clostridium difficile-associated diarrhoea) in care homes. Health Technology Assessment. 2014;18:1–84.
- 22. Biros M. Capacity, Vulnerability, and Informed Consent for Research. The Journal of Law, Medicine & Ethics. 2018;46:72–8.
- 23. HMSO, London. Mental Capacity Act 2005.
- 24. Department of Health E, Welfare, National Commission for the Protection of Human Subjects of B, Behavioral R. The Belmont Report. Ethical principles and guidelines for the protection of human subjects of research. Journal of the American College of Dentists. 2014;81:4–13.
- 25. International Ethical Guidelines for Health-related Research Involving Humans. Geneva: Council for International Organizations of Medical Sciences (CIOMS) in collaboration with the World Health Organization (WHO); 2016. www.cioms.ch,. Accessed 21 Feb 2019.
- 26. European Parliament and Council of the European Union. Regulation (EU) No 536/2014. Official Journal of the European Union. 2014.
- 27. Huls H, Abdulahad S, Mackus M, van de Loo AJAE, Roehrs T, Roth T, et al. Inclusion and Exclusion Criteria of Clinical Trials for Insomnia. J Clin Med. 2018;7. doi:10.3390/jcm7080206.
- 28. Spaul SW, Hudson R, Harvey C, Macdonald H, Perez J. Exclusion criterion: learning disability. The Lancet. 2020;395:e29.

- 29. The GenOSept National Coordinators, European Society of Intensive Care Medicine, Tridente A, Holloway PAH, Hutton P, Gordon AC, Mills GH, et al. Methodological challenges in European ethics approvals for a genetic epidemiology study in critically ill patients: the GenOSept experience. BMC Medical Ethics. 2019;20. doi:10.1186/s12910-019-0370-1.
- 30. Gong MN, Winkel G, Rhodes R, Richardson LD, Silverstein JH. Surrogate consent for research involving adults with impaired decision making: Survey of Institutional Review Board practices. Critical care medicine. 2010;38:2146–54.
- 31. Wynn S. Decisions by Surrogates: An Overview of Surrogate Consent Laws in the United States. Bifocal: Journal of ABA Commission on Law and Aging. 2014;36. https://www.americanbar.org/groups/law_aging/publications/bifocal/vol_36/issue_1_october2014/default_surrogate_consent_statutes/. Accessed 26 Feb 2020.
- 32. Yadav KN, Gabler NB, Cooney E, Kent S, Kim J, Herbst N, et al. Approximately One In Three US Adults Completes Any Type Of Advance Directive For End-Of-Life Care. Health Aff (Millwood). 2017;36:1244–51.
- 33. Ries NM, Thompson KA, Lowe M. Including People with Dementia in Research: An Analysis of Australian Ethical and Legal Rules and Recommendations for Reform. Bioethical Inquiry. 2017;14:359–74.
- 34. Eichler H-G, Cavaleri M, Enzmann H, Scotti F, Sepodes B, Sweeney F, et al. Clinical trials for Covid-19: can we better use the short window of opportunity? Clinical Pharmacology & Therapeutics. doi:10.1002/cpt.1891.
- 35. Shepherd V, Wood F, Griffith R, Sheehan M, Hood K. Research involving adults lacking capacity to consent: a content analysis of participant information sheets for consultees and legal representatives in England and Wales. Trials. 2019;20:233.
- 36. Dixon-Woods M, Angell EL. Research involving adults who lack capacity: how have research ethics committees interpreted the requirements? Journal of Medical Ethics. 2009;35:377–81.
- 37. Shepherd V, Griffith R, Sheehan M, Wood F, Hood K. Healthcare professionals' understanding of the legislation governing research involving adults lacking mental capacity in England and Wales: a national survey. Journal of Medical Ethics. 2018.
- 38. Bolcic-jankovic D, Clarridge BR, Leblanc JL, Mahmood RS, Roman AM, Freeman BD. Exploring Determinants of Surrogate Decision-Maker Confidence: An Example From the ICU. Journal of Empirical Research on Human Research Ethics. 2014;9:76–85.
- 39. Bryant J, Skolarus LE, Smith B, Adelman EE, Meurer WJ. The accuracy of surrogate decision makers: Informed consent in hypothetical acute stroke scenarios. BMC Emergency Medicine. 2013;13 (1) (no.
- 40. Ciroldi M, Cariou A, Adrie C, Annane D, Castelain V, Cohen Y, et al. Ability of family members to predict patient's consent to critical care research. Intensive Care Medicine. 2007;33:807–13.
- 41. Coppolino M, Ackerson L. Do surrogate decision makers provide accurate consent for intensive care research? Chest. 2001;119:603–12.
- 42. Newman JT, Smart A, Tyler RR, Williams A, Moss M, Newman JT, et al. Surrogate and patient discrepancy regarding consent for critical care research. Critical Care Medicine. 2012;40:2590.

- 43. Stocking CB, Hougham GW, Danner DD, Patterson MB, Whitehouse PJ, Sachs GA. Speaking of research advance directives Planning for future research participation. Neurology. 2006;66:1361–6.
- 44. Johansson M, Broström L. Turning failures into successes: A methodological shortcoming in empirical research on surrogate accuracy. Theoretical Medicine and Bioethics. 2008;29:17–26.
- 45. Kohn NA, Kohn N. Matched Preferences and Values: A New Approach to Selecting Legal Surrogates. San Diego Law Review, Vol 52, 2015. 2015;1.
- 46. Shepherd V, Hood K, Sheehan M, Griffith R, Jordan A, Wood F. Ethical understandings of proxy decision making for research involving adults lacking capacity: A systematic review (framework synthesis) of empirical research. AJOB Empirical Bioethics. 2018;:1–20.
- 47. Berger J, Majerovitz S. Do elderly persons' concerns for family burden influence their preferences for future participation in dementia research. Journal of Clinical Ethics. 2005;16:108–15.
- 48. Del Giudice A, Plaum J, Maloney E, Kasner SE, Le Roux PD, Baren JM. Who Will Consent to Emergency Treatment Trials for Subarachnoid Hemorrhage? Academic Emergency Medicine. 2009;16:309–15.
- 49. Sachs GA. Advance consent for dementia research. Alzheimer Disease & Associated Disorders. 1994;8:19–27.
- 50. Kim SYH, Kim HM, Ryan KA, Appelbaum PS, Knopman DS, Damschroder L, et al. How Important Is "Accuracy" of Surrogate Decision-Making for Research Participation? PLoS ONE. 2013;8 (1) (no.
- 51. Mehta S, Pelletier FQ, Brown M, Ethier C, Wells D, Burry L, et al. Why substitute decision makers provide or decline consent for ICU research studies: a questionnaire study. Intensive Care Medicine. 2012;38:47–54.
- 52. Shepherd V, Hood K, Sheehan M, Griffith R, Wood F. 'It's a tough decision': A qualitative study of proxy decision-making for research involving adults who lack capacity to consent in UK. Age and Ageing. 2019;:1–7.
- 53. Kim SYH, Uhlmann RA, Appelbaum PS, Knopman DS, Kim HM, Damschroder L, et al. Deliberative assessment of surrogate consent in dementia research. Alzheimer's and Dementia. 2010;6:342–50.
- 54. Chenaud C, Merlani P, Verdon M, Ricou B. Who should consent for research in adult intensive care? Preferences of patients and their relatives: a pilot study. J Med Ethics. 2009;35:709–12.
- 55. Scales DC, Smith OM, Pinto R, Barrett KA, Friedrich JO, Lazar NM, et al. Patients' preferences for enrolment into critical-care trials. Intensive Care Medicine. 2009;35:1703–12.
- 56. Heyland DK, Cook DJ, Rocker GM, Dodek PM, Kutsogiannis DJ, Peters S, et al. Decision-making in the ICU: perspectives of the substitute decision-maker. Intensive Care Med. 2003;29:75–82.
- 57. Grap MJ, Munro CL. Subject recruitment in critical care nursing research: a complex task in a complex environment. Heart Lung. 2003;32:162–8.
- 58. Larkin ME, Beauharnais CC, Magyar K, Macey L, Grennan KB, Boykin EE, et al. Obtaining surrogate consent for a minimal-risk research study in the intensive care unit setting. Clinical Trials. 2013;10:93–6.

- 59. Turnbull AE, Hashem MD, Rabiee A, To A, Chessare CM, Needham DM. Evaluation of a strategy for enrolling the families of critically ill patients in research using limited human resources. PLOS ONE. 2017;12:e0177741.
- 60. Shepherd V, Davies J. Conducting a Randomized Controlled Trial in Care Homes: The Challenges of Recruiting Residents Who Lack Capacity to Consent. SAGE Research Methods Cases: Medicine and Health. 2020. doi:10.4135/9781529726626.
- 61. U.S. Department of Health and Human Services, Food and Drug Administration. FDA Guidance on Conduct of Clinical Trials of Medical Products during COVID-19 Public Health Emergency. 2020. https://www.fda.gov/media/136238/download.
- 62. Ries N, Mansfield E, Sanson-Fisher R. Planning Ahead for Dementia Research Participation: Insights from a Survey of Older Australians and Implications for Ethics, Law and Practice. J Bioeth Inq. 2019;16:415–29.
- 63. Karlawish J, Rubright J, Casarett D, Cary M, Ten Have T, Sankar P. Older Adults' Attitudes Toward Enrollment of Non-competent Subjects Participating in Alzheimer's Research. American Journal of Psychiatry. 2009;166:182–8.
- 64. Shepherd V, Griffith R, Hood K, Sheehan M, Wood F. "There's more to life than money and health": Family caregivers' views on the role of Power of Attorney in proxy decisions about research participation for people living with dementia. Dementia (London). 2019;:1471301219884426.
- 65. Iverson E, Celious A, Kennedy CR, Shehane E, Eastman A, Warren V, et al. Real-time Perspectives of Surrogate Decision-Makers Regarding Critical Illness Research: findings of focus group participants. Chest. 2012;142:1433–9.
- 66. Sugarman J, Cain C, Wallace R, Welsh-Bohmer KA. How proxies make decisions about research for patients with Alzheimer's disease. Journal of the American Geriatrics Society. 2001;49:1110–9.
- 67. Azoulay E, Pochard F, Kentish-Barnes N, Chevret S, Aboab J, Adrie C, et al. Risk of Post-traumatic Stress Symptoms in Family Members of Intensive Care Unit Patients. American Journal of Respiratory and Critical Care Medicine. 2005;171:987–94.
- 68. Shelton AK, Freeman BD, Fish AF, Bachman JA, Richardson LI. A Computer-Based Education Intervention to Enhance Surrogates' Informed Consent for Genomics Research. American Journal of Critical Care. 2015;24:148–55.
- 69. Shepherd V, Hood K, Sheehan M, Griffith R, Wood F. Development of a complex intervention to support informed decision-making by family members of adults who lack capacity to consent to trials. Trials. 20 Suppl 1 PS2C-O3:579.
- 70. Resnik DB, Miller F. The ethics of sham surgery on research subjects with cognitive impairments that affect decision-making capacity. Contemporary Clinical Trials. 2010;31:407–10.
- 71. Werner P, Schicktanz S. Practical and Ethical Aspects of Advance Research Directives for Research on Healthy Aging: German and Israeli Professionals' Perspectives. Front Med (Lausanne). 2018;5. doi:10.3389/fmed.2018.00081.
- 72. Booth MG. Informed Consent in Emergency Research: A Contradiction in Terms. Sci Eng Ethics. 2007;13:351–9.

- 73. Lemaire F, European Society of Intensive Care Medicine. European Society of Intensive Care Medicine statement. Clinical research in intensive care unit. Intensive Care Med. 2002;28:1218–9.
- 74. Dutton RP, Stansbury LG, Hemlock B, Hess JR, Scalea TM. Impediments to Obtaining Informed Consent for Clinical Research in Trauma Patients. Journal of Trauma and Acute Care Surgery. 2008;64:1106–1112.
- 75. Schandelmaier S, von Elm E, You JJ, Blümle A, Tomonaga Y, Lamontagne F, et al. Premature Discontinuation of Randomized Trials in Critical and Emergency Care: A Retrospective Cohort Study. Crit Care Med. 2016;44:130–7.
- 76. Roberts I, Prieto-Merino D, Shakur H, Chalmers I, Nicholl J. Effect of consent rituals on mortality in emergency care research. The Lancet. 2011;377:1071–2.
- 77. Lecouturier J, Rodgers H, Ford GA, Rapley T, Stobbart L, Louw SJ, et al. Clinical research without consent in adults in the emergency setting: a review of patient and public views. BMC Medical Ethics. 2008;9:9.
- 78. ICH-GCP. Guideline for Good Clinical Practice E6(R2) by the International Conference on Harmonisation of Technical Requirements for Registration of Pharmaceuticals for Human Use. 2016.
- 79. Guidance for Institutional Review Boards, Clinical Investigators, and Sponsors. Exception from Informed Consent Requirements for Emergency Research. U.S. Department of Health and Human Services Food and Drug Administration; 2013. https://www.fda.gov/media/80554/download. Accessed 28 Feb 2020.
- 80. Klein L, Moore J, Biros M. A 20-year Review: The Use of Exception From Informed Consent and Waiver of Informed Consent in Emergency Research. Academic Emergency Medicine. 2018;25:1169–77.
- 81. European Commission. Clinical Trials Directive 2001/20/EC. 2001. https://ec.europa.eu/health/human-use/clinical-trials/directive en. Accessed 26 Jan 2018.
- 82. Robinson K, Andrews PJD. '(More) trials and tribulations': the effect of the EU directive on clinical trials in intensive care and emergency medicine, five years after its implementation. Journal of Medical Ethics. 2010;36:322–5.
- 83. The Medicines for Human Use (Clinical Trials) Amendment (No.2) Regulations 2006. http://www.legislation.gov.uk/uksi/2006/2984/contents/made. Accessed 28 Feb 2020.
- 84. Kompanje EJO, Maas AIR, Menon DK, Kesecioglu J. Medical research in emergency research in the European Union member states: tensions between theory and practice. Intensive Care Med. 2014;40:496–503.
- 85. Mentzelopoulos SD, Mantzanas M, van Belle G, Nichol G. Evolution of European Union legislation on emergency research. Resuscitation. 2015;91:84–91.
- 86. Berger BJ. Minimum Risk and HEAT-PPCI: Innovative Ideas for Informed Consent in Emergency Medical Research. Annals of Emergency Medicine. 2014;64:A17–9.
- 87. Davies H, Shakur H, Padkin A, Roberts I, Slowther A-M, Perkins GD. Guide to the design and review of emergency research when it is proposed that consent and consultation be waived. Emerg Med J. 2014;31:794–5.